Author’s response to reviews

Title: A qualitative study on barriers in the prevention of anaemia during pregnancy in Public Health Centres: perceptions of Indonesian nurse-midwives.

Authors:

Widyawati - Widyawati - (a.widyawati@radboudumc.nl)
Suze - Jans - (suzejans@dds.nl)
Sutarti - Utomo - (hera09@yahoo.com)
Jeroen van Dillen - (jeroen.vandillen1@radboudumc.nl)
Toine Lagro Janssen - (toine.lagro@radboudumc.nl)

Version: 6 Date: 5 February 2015

Author’s response to reviews: see over
Date : February 6, 2015
To Editor : BMC Pregnancy & Childbirth

Dear Editor,

First of all we would like to say thank you to the editor and reviewers for the positive response and the decision on our manuscript publication. With this letter we would like to re-submit our manuscript revision entitled, “A qualitative study on barriers in the prevention of anaemia during pregnancy in Public Health Centres: perceptions of Indonesian nurse-midwives,” for the next process of publication in BMC Pregnancy & Childbirth.

Regarding to the editors comment on our manuscript, here is the revisions:

1. Please remove all author suffixes from the title page
   Answer: Yes, we removed all author suffixes from the title page

2. Please move all the tables to the end of the manuscript according to journal
   Answer: Yes, we moved all the tables to the end of the manuscript

3. Please change Limitation and strengths section so that it is a subsection in the Discussion instead
   Answer: Yes, we changed Limitation and strengths as a subsection in the Discussion

4. Please remove the funding section and include this information in the
   Acknowledgement instead
   Answer: Yes, we removed the information of the funding in the Acknowledgement.

Thank you for giving us a chance to publish our manuscript in your journal. We appreciate your time and we are looking forward to your response

Best regards,

Widyawati, BSN., MPH.
PhD candidate
Department of Primary and Community Care, Gender & Women’s Health, Radboud University Medical Centre Nijmegen, The Netherlands
School of Nursing, Faculty of Medicine, Universitas Gadjah Mada, Yogyakarta – Indonesia.

Email: A.Widyawati@radbourumc.nl, widyawati_ugm@yahoo.com